Effect of Bariatric Surgery on CKD Risk

Allon N. Friedman,¹ Abdus S. Wahed,² Junyao Wang,² Anita P. Courcoulas,³ Gregory Dakin,⁴ Marcelo W. Hinojosa,⁵ Paul L. Kimmel,⁶ James E. Mitchell,⁷ Alfons Pomp,⁵ Walter J. Pories,⁸ Jonathan Q. Purnell,⁹ Carel le Roux,¹⁰ Konstantinos Spaniolas,¹¹ Kristine J. Steffen,¹² Richard Thirlby,¹³ and Bruce Wolfe¹⁴

¹Department of Medicine, Indiana University School of Medicine, Indianapolis Indiana; ²Department of Biostatistics, University of Pittsburgh Graduate School of Public Health, Pittsburgh, Pennsylvania; ³Department of Surgery, University of Pittsburgh Medical Center, Pittsburgh, Pennsylvania; ⁴Department of Surgery, Weill Cornell Medical College, New York, New York; ⁵Department of Surgery, University of California, Irvine Medical Center, Irvine, California; ⁶Division of Kidney, Urologic, and Hematologic diseases, National Institutes of Diabetes and Digestive and Kidney Diseases, National Institutes of Health, Bethesda, Maryland; ⁷Neuropsychiatric Research Institute, Fargo, North Dakota; ⁸Department of Surgery, Brody School of Medicine, Greenville, North Carolina; ⁹Department of Medicine, Oregon Health Sciences University, Portland, Oregon; ¹⁰Department of Pathology, University College Dublin, Ireland; ¹¹Department of Surgery, Stony Brook Medicine, Stony Brook, New York; ¹²Department of Pharmaceutical Sciences, North Dakota State University College of Health Professions, Fargo, North Dakota; ¹³Department of Surgery, Virginia Mason Medical Center, Seattle, Washington; and ¹⁴Department of Surgery, Oregon Health Sciences University, Portland, Oregon

ABSTRACT

Obesity is linked to the development and progression of CKD, but whether bariatric surgery protects against CKD is poorly understood. We, therefore, examined whether bariatric surgery influences CKD risk. The study included 2144 adults who underwent bariatric surgery from March of 2006 to April of 2009 and participated in the Longitudinal Assessment of Bariatric Surgery-2 Study cohort. The primary outcome was CKD risk categories as assessed by the Kidney Disease Improving Global Outcomes (KDIGO) consortium criteria using a combination of eGFR and albuminuria. Patients were 79% women and 87% white, with a median age of 46 years old. Improvements were observed in CKD risk at 1 and 7 years after surgery in patients with moderate baseline CKD risk (63% and 53%, respectively), high baseline risk (78% and 56%, respectively), and very high baseline risk (59% and 23%, respectively). The proportion of patients whose CKD risk worsened was $\leq 10\%$; five patients developed ESRD. Sensitivity analyses using year 1 as baseline to minimize the effect of weight loss on serum creatinine and differing eGFR equations offered qualitatively similar results. Treatment with bariatric surgery associated with an improvement in CKD risk categories in a large proportion of patients for up to 7 years, especially in those with moderate and high baseline risk. These findings support consideration of CKD risk in evaluation for bariatric surgery as a treatment for high-risk obese patients with CKD.

J Am Soc Nephrol 29: 1289-1300, 2018. doi: https://doi.org/10.1681/ASN.2017060707

Bariatric surgery is the most effective and sustained of all weight reduction strategies and results in improvements in a variety of disease states,

Copyright © 2018 by the American Society of Nephrology

including diabetes and hypertension.^{1–5} A related but relatively understudied area involves the effect of bariatric surgery on CKD.⁶ Interest in this topic has arisen because obesity is linked either directly or through intermediate diseases, like type 2 diabetes and hypertension, with the development and progression of CKD.^{7–10} This problem is compounded by the fact that nearly one in every two individuals with CKD in the United States is obese, a proportion that continues to rise.¹¹

Received June 29, 2017. Accepted December 15, 2017.

Published online ahead of print. Publication date available at www.jasn.org.

Correspondence: Dr. Allon N. Friedman, 550 University Boulevard, Suite 6100, Indianapolis, IN 46202. Email: allfried@iu.edu

However, although bariatric surgery has been shown in people with obesity and normal kidney function to lower proteinuria and reverse glomerular hyperfiltration,^{12–14} the implications of these findings on risk for CKD are unclear. The few studies of bariatric surgery in patients with existing CKD suggested benefits but were limited by small sample size, short follow-up periods, or estimations of GFR that did not use the most accurate equation.^{15–18} In addition, previous studies did not control for the possibility that observed improvements in serum creatinine and eGFR, the two most commonly used markers of kidney filtration, were because of loss of muscle mass from weight loss rather than protective effects on the kidney.

The aims of our study were to assess how bariatric surgery influences CKD risk, identify patient factors associated with higher risk, and control for the distorting effects that weight loss may have on serum creatinine. The study was performed in a large, prospective, multicenter cohort that had regular follow-up.

RESULTS

Patient Characteristics

Baseline characteristics of the 2144 study participants are shown in Table 1. Seventy-nine percent of the study cohort were women, 87% were white, and the median age was 46 years old. Roux-en-Y gastric bypass was performed in 71% of participants, and laparoscopic adjustable banding was performed in 24% of participants. CKD risk categories at baseline were the following: 83.4% low, 11.9% moderate, 3.4% high, and 1.4% very high.

Factors Associated with CKD Risk Categories at Baseline

Factors in the univariable analysis that were significantly associated with a high or very high CKD risk at baseline are shown in Supplemental Table 1. After adjustment for other variables in the model, factors significantly and independently associated with high or very high CKD risk at baseline included age (per 5-year increase; 1.44; 95% confidence interval [95% CI], 1.27 to 1.62), men (1.84; 95% CI, 1.15 to 2.94), body mass index (BMI; per 5-kg/m² increase; 1.24; 95% CI, 1.08 to 1.43), and hemoglobin A1c (HbA1c; per percentage; 1.79; 95% CI, 1.56 to 2.05).

Effect of Bariatric Surgery on CKD Risk Categories over Time

Participants with accessible information on their CKD risk category at baseline and at least one follow-up visit and who underwent laparoscopic adjustable banding gastric bypass or Roux-en-Y gastric bypass were included in the longitudinal analysis (n=1807) (Supplemental Figure 1). Study participants were followed for up to 7 years after bariatric surgery. Among the 2144 participants for whom CKD risk category data were

Significance Statement

Obesity is linked to the development and progression of CKD, but whether bariatric surgery protects against CKD is not known. In this observational study of a bariatric surgery cohort, CKD risk was determined using KDIGO categories at baseline, yearly for 5 years, and in a subset for 7 years. In a high proportion of the patients with moderate or high baseline risk, bariatric surgery was associated with an improvement in the assigned CKD risk category. These findings support consideration of CKD risk in evaluation for bariatric surgery and further study of bariatric surgery as a treatment for high-risk patients with CKD.

available at baseline, 1449 participants had such data available at year 1, 1236 participants had data available at year 2, 1247 participants had data available at year 3, 1216 participants had data available at year 4, 1243 participants had data available at year 5, and 824 participants had data available at year 7. Figure 1 shows trends in CKD risk categories over time categorized by level of baseline risk. In patients with low CKD risk at baseline, the risk worsened on average in a relatively small proportion (4%-9%) of patients (i.e., moved to a higher-risk category) between baseline and years 1-7. No patients in the low-risk group improved their risk because they were already in the lowest-risk category. In the moderate baseline CKD risk group, a large proportion of individuals improved over time. Over the first year, approximately 63% improved their risk category (i.e., from moderate to low risk), and by 7 years, 53% of patients reduced their risk category compared with a much smaller proportion that worsened (approximately 5%-8% over 7 years). In the high CKD risk group at baseline, similar patterns were observed, with 78% improving their risk category by year 1 and 56% improving their risk category by year 7, with approximately 3%-10% worsening during the same time. In the very high CKD risk group at baseline, 59% improved their risk category by year 1, and 23% had improvement by year 7. No persons in the very high CKD risk category worsened their risk profile because they were already in the highest-risk category. Five patients developed ESRD during the follow-up period and were included in the very high-risk category. In a sensitivity analysis, these results did not qualitatively change after creatinine-based eGFR (eGFR_{cr}) or cystatin C-based eGFR (eGFR_{cys}) was substituted for creatinine- and cystatin C-based eGFR (eGFR_{cr-cys}) to estimate risk^{19,20} (Supplemental Figures 2 and 3). Separate median and mean data for eGFR_{cr-cys} and albumin-to-creatinine (ACR) are shown in Table 2, Table 3, Table 4, and Table 5 (with slopes in Supplemental Figures 4 and 5). Median eGFR using eGFR_{cr-cvs} peaked by year 2 and slowly declined thereafter in the overall cohort and each CKD risk subgroup. ACR levels remained stable in the low-risk group, dropped dramatically in the moderate- and high-risk groups, and ultimately worsened in the very high-risk group. Supplemental Tables 2–5 show mean and median eGFR changes using eGFR_{cr} or eGFR_{cys}.

Table 1. Baseline characteristics by CKD risk category

Variables	Low Risk	Moderate Risk	High Risk	Very High Risk
No. (%)	1788 (83.4)	254 (11.9)	73 (3.4)	29 (1.4)
Renal parameters				
eGFR _{cr-cvs} , ml/min per 1.73 m ² , median (IQR)	100 (87–111)	92 (66–107)	63 (43–98)	36 (27–41)
eGFR _{cr} , ml/min per 1.73 m ² , median (IQR)	105 (93–115)	99 (78–113)	74 (50–102)	38 (31–48)
eGFR _{cvs} , ml/min per 1.73 m ² , median (IQR)	96 (81–109)	84 (58–106)	58 (40–88)	31 (26–41)
ACR, mg/g, median (IQR)	6.0 (4.2–9.6)	48.0 (32.1–96.7)	325.5 (14.2–495.5)	345.4 (13.4–1185.7)
Serum creatinine, mg/dl, median (IQR)	0.71 (0.63–0.82)	0.80 (0.67-0.95)	1.01 (0.72–1.28)	1.78 (1.37-2.07)
Serum cystatin C, mg/L, median (IQR)	0.86 (0.76–0.97)	0.95 (0.79–1.22)	1.24 (0.89–1.61)	1.89 (1.62–2.22)
Demographic characteristics				
Age, yr, median (IQR)	45 (36–53)	48 (39–56)	54 (43–60)	57 (52–63)
Sex, no. (%)				
Men	326 (18.2)	85 (33.5)	24 (32.9)	13 (44.8)
Women	1462 (81.8)	169 (66.5)	49 (67.1)	16 (55.2)
Race, no. (%)				
White	1552 (86.8)	221 (87.0)	61 (83.6)	26 (89.7)
Black	185 (10.3)	25 (9.8)	5 (6.8)	2 (6.9)
Other	51 (2.9)	8 (3.1)	7 (9.6)	1 (3.4)
Annual household income, no. (%)	1632	233	65	26)
<\$25,000	271 (16.6)	50 (21.5)	15 (23.1)	4 (15.4)
\$25,000-\$49,999	439 (26.9)	51 (21.9)	15 (23.1)	9 (34.6)
\$50,000-\$74,999	400 (24.5)	58 (24.9)	14 (21.5)	5 (19.2)
\$75,000-\$99,999	262 (16.1)	35 (15.0)	13 (20.0)	4 (15.4)
≥\$100,000	260 (15.9)	39 (16.7)	8 (12.3)	4 (15.4)
Education, no. (%)	1672	237	69	28
High school or less	378 (22.6)	49 (20.7)	19 (27.5)	6 (21.4)
Some college	692 (41.4)	98 (41.4)	29 (42.0)	12 (42.9)
College degree or higher	602 (36.0)	90 (38.0)	21 (30.4)	10 (35.7)
Medical insurance type, no. (%)	1670	237	69	28
No insurance	19 (1.1)	2 (0.8)	0 (0)	0 (0)
Medicaid	146 (8.7)	33 (13.9)	7 (10.1)	1 (3.6)
Medicare	159 (9.5)	39 (16.5)	16 (23.2)	6 (21.4)
Tricare	44 (2.6)	9 (3.8)	5 (7.2)	3 (10.7)
Private	1148 (68.7)	129 (54.4)	38 (55.1)	17 (60.7)
Other/unknown insurance type	154 (9.2)	25 (10.5)	3 (4.3)	1 (3.6)
Anthropometric measures				
BMI, kg/m², median (IQR)	45.6 (41.6–51.0)	46.8 (42.7–52.8)	46.0 (42.1–52.6)	48.1 (41.6–51.9)
Weight, kg, median (IQR)	128 (114–145)	136 (117–156)	133 (115–155)	135 (119–152)
Waist circumference, cm				
n	1695	239	69	26
Median (IQR)	130 (120–141)	137 (126–148)	138 (127–149)	140 (129–148)
Percentage body fat				
n	1507	213	60	21
Median (IQR)	51.3 (48.5–53.8)	50.8 (46.8–54)	50.7 (46.3–53.3)	48.9 (38.3–53.9)
Glycemic/insulin data				
Diabetes				
n	1752	249	72	29
No. (%)	486 (27.7)	128 (51.4)	53 (73.6)	24 (82.8)
Insulin, μ IU/ml				
n	1523	222	65	26
Median (IQR)	18.7 (12.7–28.9)	21.9 (16–33.2)	24.3 (16.4–39.7)	19.2 (12–38.7)
Fasting glucose, mg/dl				
n	1516	221	65	26
Median (IQR)	97 (89–110)	106 (94–137)	114 (100–158)	112 (89–186)
HOMA_IR ^a				
n	1515	221	65	26
Median (IQR)	4.6 (2.9–7.6)	6.3 (4.2–10.7)	8.0 (4.9–14.6)	7.2 (3.8–10.4)

Table 1. Continued

Variables	Low Risk	Moderate Risk	High Risk	Very High Risk
HbA1c, %				
n	1777	252	73	29
Median (IQR)	5.5 (5.2–6.1)	6 (5.5–7.3)	6.8 (5.9–8.4)	6.6 (5.7–7.9)
BP				
Hypertension				
n	1730	247	71	29
No. (%)	1112 (64.3)	194 (78.5)	61 (85.9)	28 (96.6)
SBP				
n	1750	251	72	29
Median (IQR)	128 (120–139)	132 (120–142)	130 (119–140)	131 (121–147)
DBP				
n	1750	251	72	29
Median (IQR)	80 (73–85)	80 (71–86)	72 (65–81)	77 (68–81)
Habits				
Smoking, no. (%)				
Never	1011 (56.5)	142 (55.9)	33 (45.2)	15 (51.7)
Current	80 (4.5)	9 (3.5)	4 (5.5)	1 (3.4)
Former	697 (39.0)	103 (40.6)	36 (49.3)	13 (44.8)
Use of ACE inhibitor or ARB medications, no. (%)	566 (31.7)	124 (48.8)	44 (60.3)	19 (65.5)
Bariatric surgery type, no. (%)				
Roux-en-Y gastric bypass	1287 (72.0)	170 (66.9)	53 (72.6)	20 (69)
Laparoscopic adjustable band	435 (24.3)	68 (26.8)	14 (19.2)	6 (20.7)
Sleeve gastrectomy	34 (1.9)	12 (4.7)	4 (5.5)	2 (6.9)
BPDS	15 (0.8)	2 (0.8)	0 (0)	0 (0)
Banded gastric bypass	17 (1)	2 (0.8)	2 (2.7)	1 (3.4)

SI conversion factors are as follows. To convert serum creatinine to micromoles per liter, multiply by 76.26. To convert insulin to picomoles per liter, multiply by 6.945. To convert glucose to millimoles per liter, multiply by 0.0555. To convert HbA1c to proportion of total Hb, multiply by 0.01. ACE inhibitor can indicate ACE inhibitor, ACE inhibitor/calcium channel blocker combination, or ACE inhibitor/thiazide combination. ARB can indicate angiotensin II receptor antagonist/thiazide combination. IQR, interquartile range; HOMA_IR, insulin resistance by homeostatic model assessment; SBP, systolic BP; DBP, diastolic BP; BPDS, biliopancreatic diversion with duodenal switch.

^aHOMA_IR was calculated as [insulin (microunits per milliliter) \times glucose (milligrams per deciliter)]/405.

Because weight loss after bariatric surgery leads to loss of muscle in addition to fat²¹ and because muscle is the major source of endogenous creatinine,²² we performed a sensitivity analysis in which the baseline was redefined as year 1, which was when weight loss in most patients plateaued⁵ (Figure 2). This was done to minimize the likelihood that any observed improvement in CKD risk categories could be related to muscle loss. Only 5% of individuals with low CKD risk at baseline went on to develop higher risk in year 2, and only 10% had higher risk in year 7. A substantial proportion of persons improved their CKD risk in the moderate (51% in year 2 and 35% by year 7) and high (58% in year 2 and 36% by year 7) baseline risk groups, with only a modest proportion developing higher CKD risk (moderate baseline risk group: 10% at year 2 and 19% at year 7; high baseline risk: 8% in year 2 and 20% in year 7). Patients with very high baseline CKD risk had a lower proportion of individuals whose risk improved over time (30% in year 2 and 36% by year 7). Results were not qualitatively different using eGFR_{cr} or eGFR_{cys} (Supplemental Figures 6 and 7).

In univariable analysis, baseline characteristics significantly associated with developing moderate, high, or very high CKD risk during the follow-up period are shown in Supplemental Table 6. Of note, CKD risk over time was dependent on age (age and visit interaction P < 0.001) (Supplemental Figure 8, Supplemental Table 7). For example, a 38-year old participant had a 52% lower risk of having moderate, high, or very high CKD risk at year 3 compared with baseline (relative risk [RR], 0.51; 95% CI, 0.37 to 0.69) compared with only a 28% lower risk (RR, 0.72; 95% CI, 0.63 to 0.81) for a 55-year-old participant.

In the multivariable model (Table 6), variables significantly associated with a high or very high CKD risk included baseline BMI (RR, 1.12; 95% CI, 1.06 to 1.20 per 5-kg/m² increase), HbA1c (1.25; 95% CI, 1.19 to 1.31 per 1% increase), use of angiotensin-converting enzyme (ACE) inhibitors or angiotensin receptor blockers (ARBs; 1.38; 95% CI, 1.13 to 1.68), men (RR, 1.34; 95% CI, 1.10 to 1.63), lack of private medical insurance (1.34; 95% CI, 1.11 to 1.61), and less weight lost (1.05; 95% CI, 1.01 to 1.09 per 5% lower). Similar to the unadjusted model, CKD risk categories over time were age dependent (Supplemental Table 8). Replacing HbA1c in the model with presence of diabetes (RR, 1.85; 95% CI, 1.52 to 2.25) or percentage weight loss from baseline with reduction in waist



Figure 1. Trend of overall improvement in CKD risk categories over 7-year period after bariatric surgery. Predicted proportions are obtained from generalized linear mixed models stratified by baseline prognostic CKD risk category and include visit and missing related baseline variables, age, race, and smoking status. The numbers above each bar represent the total numbers of participants with accessible data at each time point. The total number at baseline is shown on the far right of each row. BL, baseline.

Table 2.	Median eGF	R _{cr-cys} a over a	а 7-уе	ar follow-u	p period	by	baseline	CKD	risk	category	y
----------	------------	------------------------------	--------	-------------	----------	----	----------	-----	------	----------	---

		Overall		Low Risk		Moderate Risk		High Risk		Very High Risk	
	N ^c	Median (IQR)	N ^d	Median (IQR)	N ^d	Median (IQR)	N ^d	Median (IQR)	N ^d	Median (IQR)	
Baseline (2453)	2330	98 (84–110)	1788	100 (87–111)	254	92 (66–107)	73	63 (43–98)	29	36 (27–41)	
12 mo (2397)	1696	103 (89–114)	1260	105 (93–115)	178	96 (74–110)	57	75 (59–101)	17	43 (39–60)	
24 mo (2360)	1453	104 (90–115)	1076	106 (95–117)	159	98 (78–111)	51	77 (61–103)	17	48 (37–58)	
36 mo (2372)	1434	103 (89–114)	1067	105 (93–115)	156	98 (80–113)	50	73 (62–105)	16	45 (29–56)	
48 mo (2373)	1394	101 (85–111)	1045	103 (90–113)	141	96 (76–108)	52	69 (59–93)	16	36 (17–50)	
60 mo (2378)	1455	98 (81–110)	1070	100 (86–111)	161	86 (66–105)	55	68 (59–93)	13	38 (13–44)	
84 mo (2360)	937	91 (78–105)	716	94 (82–106)	100	81 (64–104)	31	69 (52–90)	8	25 (10-44)	

The difference between the N values in the first and second columns is the number of participants who do not have data to calculate eGFR_{cr-cvs}, whereas the difference between the N in the second column and the sum of the N values in columns 3-6 is the number of participants who were missing a CKD risk category because of missing ACR data. IQR, interquartile range.

^aMilliliters per minute per 1.73 m².

^bTotal number of available participants at each time point. ^cNumber of participants with sufficient data to calculate eGFR_{cr-cys}.

^dNumber of participants with sufficient data to assess CKD risk category using eGFR_{cr-cys} and ACR.

circumference from baseline (RR, 1.05; 95% CI, 1.02 to 1.08 per 5-cm decrease) did not qualitatively change results. Importantly, presence of hypertension, baseline systolic BP, and type of bariatric surgery were not significantly associated with CKD risk categories during follow-up.

DISCUSSION

The study's major finding was that bariatric surgery resulted in lower CKD risk in a substantial proportion of patients throughout the 7-year follow-up period. Reduction in risk was most pronounced in persons with high baseline risk. However, improvements were also observed in patients with moderate and to a lesser extent, very high baseline risk.

We identified several variables that conferred higher CKD risk after bariatric surgery in addition to traditional risk factors, like men, greater adiposity, and diabetes. The higher risk associated with ACE inhibitors/ARBs may reflect the fact that their use was reserved for sicker patients in light of reports

suggesting that they offer renoprotection in people with obesity.23 Similarly, the protective association with private medical insurance could be indicative of better access to health care. What is notable is that greater weight loss and not the mechanism through which weight loss was achieved (i.e., type of bariatric surgery) was an independent predictor of reduced CKD risk. This important and novel finding is consistent with reports in rats in which equivalent weight loss from surgical or medical interventions led to similar histologic improvements in the kidney.24

In comparison with the previous literature,^{15–18,25} our study offers several important strengths, including a prospective multicenter design, large size, and long follow-up period. Additionally, we estimated GFR using eGFR_{cr-cys}, which we previously identified as the most accurate in the bariatric surgery population¹² and that incorporates the two major and complementary endogenous filtration markers serum creatinine and cystatin C.²⁰ Because both eGFR and albuminuria were available at regular time points, we were also able to assess CKD risk using the Kidney Disease Improving Global

Table 3. Mean eGFR_{cr-cvs^a} over a 7-year follow-up period by baseline CKD risk category

		-)								
Visit (<i>N</i>) ^b	(Overall		Low Risk		Moderate Risk		High Risk		y High Risk
	N ^c	Mean (SD)	N ^d	Mean (SD)	N ^d	Mean (SD)	N ^d	Mean (SD)	N ^d	Mean (SD)
Baseline (2453)	2330	96 (21.4)	1788	99 (16.9)	254	88 (24.6)	73	71 (29.1)	29	34 (12.3)
12 mo (2397)	1696	100 (21.3)	1260	104 (17.9)	178	92 (23.7)	57	78 (25.5)	17	48 (17.7)
24 mo (2360)	1453	102 (20.9)	1076	105 (17.2)	159	94 (21.7)	51	80 (26.4)	17	50 (21.8)
36 mo (2372)	1434	100 (21.2)	1067	104 (17.6)	156	94 (24.2)	50	79 (26.3)	16	45 (21.6)
48 mo (2373)	1394	97 (22.1)	1045	101 (18.6)	141	92 (22.6)	52	74 (24.4)	16	36 (20.2)
60 mo (2378)	1455	95 (22.1)	1070	98 (18.6)	161	86 (24.6)	55	74 (24.8)	13	33 (17.6)
84 mo (2360)	937	90 (21.8)	716	93 (18.8)	100	82 (24.6)	31	73 (24.0)	8	31 (25.1)

The difference between the N values in the first and second columns is the number of participants who do not have data to calculate eGFR_{cr-cvs}, whereas the difference between the N in the second column and the sum of the N values in columns 3-6 is the number of participants who were missing a CKD risk category because of missing ACR data.

^aMilliliters per minute per 1.73 m².

^bTotal number of available participants at each time point.

°Number of participants with sufficient data to calculate $eGFR_{cr-cys}$.

^dNumber of participants with sufficient data to assess CKD risk category using eGFR_{cr-cys} and ACR.

Table 4.	Median ACR ^a	over a 7-y	ear follow-up	period by	v baseline CKD risk ca	tegory
						~ ~

		Overall		Low Risk		Moderate Risk		High Risk	Very High Risk		
Visit (N)	N ^c	Median (IQR)	N ^d	Median (IQR)	N ^d	Median (IQR)	N ^d	Median (IQR)	N ^d	Median (IQR)	
Baseline (2453)	2192	6.8 (4.5–13.3)	1788	6.0 (4.2–9.6)	254	48.0 (32.1–96.7)	73	325.6 (14.2–495.5)	29	245.4 (13.4–1185.8)	
12 mo (2397)	1618	6.5 (4.4–10.7)	1217	5.9 (4.3–9.1)	167	11.9 (6.2–32.0)	53	21.2 (8.1–73.8)	16	237.7 (13.9–1316.6)	
24 mo (2360)	1386	6.4 (4.3–11.3)	1036	5.9 (4.1–9.3)	148	12.1 (5.9–26.4)	46	15.4 (6.9–52.8)	15	761.3 (81.6–1472.4)	
36 mo (2372)	1414	6.5 (4.4–11.9)	1053	5.8 (4.2–9.6)	151	12.1 (6.6–33.6)	49	25.1 (12.3–69.9)	16	484.2 (118.9–905.6)	
48 mo (2373)	1383	6.4 (4.3–11.8)	1035	5.9 (4.1–9.9)	138	12.3 (5.1–34.8)	50	18.4 (6.4–85.5)	17	385.5 (157.1–1056.5)	
60 mo (2378)	1424	6.6 (4.4–12.6)	1050	6.0 (4.1–10.1)	153	12.8 (5.9–32.5)	50	28.2 (10.8–89.7)	12	273.2 (74.4–1618.4)	
84 mo (2360)	931	6.9 (4.2–12.0)	706	6.1 (4.0–10.1)	101	13.5 (5.9–30.4)	29	25.7 (6.9–79.5)	9	722.2 (424.3–1014.5)	

The difference between the N values in the first and second columns is the number of participants who do not have data to calculate eGFR_{cr-cys}, whereas the difference between the N in the second column and the sum of the N values in columns 3–6 is the number of participants who were missing a CKD risk category because of missing ACR data. IQR, interquartile range.

^aMg albumin/g creatinine.

^bTotal number of available participants at each time point.

^cNumber of participants with sufficient data to calculate eGFR_{cr-cys}.

^dNumber of participants with sufficient data to assess CKD risk category using eGFR_{cr-cys} and ACR.

Outcomes (KDIGO) criteria, which was validated in over 1.5 million persons and offers more clinical information than the traditional CKD staging system on the basis of eGFR alone.²⁶ Substituting eGFR_{cr} and eGFR_{cys} for eGFR_{cr-cys} gave similar results. To account for the fact that weight loss might lower serum creatinine (and thereby, spuriously raise eGFR and improve the CKD risk profile), we performed an additional analysis using year 1 as baseline. Although a lower proportion of patients had improvement in CKD risk after surgery, results were qualitatively similar to the main analysis. This suggests that, although weight reduction–related changes in muscle mass influence eGFR to a certain extent, bariatric surgery in fact has salutary effects on kidney health.

We also found that patients in higher baseline CKD risk categories benefited from bariatric surgery. This is an important observation not only because these individuals are most likely to have progression of CKD to ESRD but also, because they are at higher surgical risk.²⁷ In fact, it is possible that our findings may have underestimated the benefits of bariatric surgery if some of the patients whose risk status remained unchanged after surgery would have naturally progressed without weight loss. Similarly, the relatively lower protective effect noted in older patients should not necessarily be interpreted as indicating that bariatric surgery is not beneficial in such patients, because their CKD risk could have been higher had surgery not been performed. Of note, although a small proportion of patients increased their CKD risk during followup, this could simply have reflected progression of their original disease. Alternatively, these patients could have suffered from complications of AKI, kidney stones, or even oxalate nephropathy, all of which have been reported after bariatric surgery.⁶ Clarification of these issues requires a control group, which our study lacked.

Our study findings reinforce previous interest in bariatric surgery as a potential treatment option for CKD.⁶ Although bariatric surgery is a costly intervention with real risks, few effective treatments are currently available to slow or stop the progression of CKD, which is associated with rates of hospitalization, rehospitalization, and death that are several times higher than those in patients without CKD.²⁸ The risks of

Table 5.	Mean ACR ^a	over a 7	7-year	follow-up	period by	y baseline CKD) risk category

		Overall		Low Risk		Moderate Risk		High Risk		Very High Risk		
	N ^c	Mean (SD)	N ^d	Mean (SD)	N ^d	Mean (SD)	N ^d	Mean (SD)	N ^d	Mean (SD)		
Baseline (2453)	2192	44.4 (291.92)	1788	7.9 (5.52)	254	67.9 (58.40)	73	429.3 (638.55)	29	1137.2 (1963.37)		
12 mo (2397)	1618	27.8 (176.96)	1217	9.7 (20.52)	167	49.3 (133.12)	53	72.2 (115.32)	16	843.2 (1108.12)		
24 mo (2360)	1386	37.1 (327.39)	1036	11.2 (31.91)	148	28.0 (39.93)	46	62.0 (121.30)	15	1379.9 (2034.14)		
36 mo (2372)	1414	27.6 (181.16)	1053	11.2 (30.89)	151	37.2 (106.68)	49	96.4 (164.50)	16	956.1 (1365.50)		
48 mo (2373)	1383	32.5 (260.25)	1035	12.2 (57.80)	138	39.6 (75.43)	50	73.7 (121.25)	17	1223.4 (1997.80)		
60 mo (2378)	1424	30.6 (221.11)	1050	11.0 (26.38)	153	46.4 (97.82)	50	116.4 (215.94)	12	1151.0 (2035.49)		
84 mo (2360)	931	35.9 (333.44)	706	11.4 (42.43)	101	38.8 (68.27)	29	95.6 (161.96)	9	1944.9 (2905.37)		

The difference between the N values in the first and second columns is the number of participants who do not have data to calculate eGFR_{cr-cys}, whereas the difference between the N in the second column and the sum of the N values in columns 3–6 is the number of participants who were missing a CKD risk category because of missing ACR data.

^aMg albumin/g creatinine.

^bTotal number of available participants at each time point.

°Number of participants with sufficient data to calculate $eGFR_{cr-cys}$.

^dNumber of participants with sufficient data to assess CKD risk category using eGFR_{cr-cys} and ACR.



Figure 2. Trend of overall improvement in CKD risk categories after bariatric surgery using year 1 post-surgery as baseline. Predicted proportions are obtained from generalized linear mixed models stratified by baseline CKD risk category and include visit and missing related baseline variables, age, race, and smoking status. The numbers above each bar represent the total numbers of participants with accessible data at each time point. The total number at year 1 is shown on the far right of each row. Y1, year 1.

 Table 6.
 Variables associated with moderate, high, or very high CKD risk over time in adjusted model

Covariate ^a	RR Ratio (95% Cl)	P Value
Presurgery characteristics		
Age (per 5-yr increase)		< 0.001
Baseline	1.13 (1.07 to 1.20)	< 0.001
Year 1	1.12 (1.04 to 1.20)	0.003
Year 2	1.20 (1.11 to 1.30)	< 0.001
Year 3	1.25 (1.15 to 1.36)	< 0.001
Year 4	1.27 (1.17 to 1.38)	< 0.001
Year 5	1.27 (1.18 to 1.36)	< 0.001
Year 7	1.31 (1.21 to 1.42)	< 0.001
Men	1.32 (1.09 to 1.59)	0.004
BMI, kg/m ² , per 5-kg/m ² increase	1.10 (1.04 to 1.17)	0.001
HbA1c, %, per 1% increase	1.24 (1.18 to 1.30)	< 0.001
SBP, mm Hg, per 5-mm Hg increase	1.02 (0.99 to 1.05)	0.12
Use of ACE inhibitor or ARB medications	1.37 (1.13 to 1.65)	0.001
No private medical insurance	1.38 (1.15 to 1.66)	0.001
Postsurgery characteristics		
Weight loss from baseline, %, per 5% decrease	1.05 (1.01 to 1.08)	0.02
Surgery type		
LAGB versus RYGB	1.17 (0.96 to 1.44)	0.13
Visit ^b		0.01
Year 1 versus baseline	0.88 (0.69 to 1.13)	0.31
Year 2 versus baseline	0.83 (0.63 to 1.10)	0.19
Year 3 versus baseline	0.77 (0.59 to 1.02)	0.07
Year 4 versus baseline	0.82 (0.63 to 1.07)	0.14
Year 5 versus baseline	1.00 (0.78 to 1.27)	0.95
Year 7 versus baseline	1.15 (0.89 to 1.50)	0.29

SBP, systolic BP; LAGB, laparoscopic adjustable banding gastric bypass; RYGB, Roux-en-Y gastric bypass.

^aA generalized linear mixed model was used to test the associations of covariates with moderate or higher prognostic risk for CKD. Independent variables with *P* values <0.40 in the univariable analyses (*i.e.*, age, sex, BMI, HbA1c, systolic BP, use of ACE inhibitor or ARB, no private medical insurance, education, annual household income, surgery type, and percentage weight loss from baseline) were considered and retained if the *P* value was <0.20 adjusted for baseline variables related to missing follow-up assessment of prognostic risk for CKD (*i.e.*, race, smoking status, and site; *n*=1623). ^bCalculated at median age of 47 years old.

morbidity and death rise further with progression to ESRD, when average annual costs can reach nearly \$100,000.²⁸ Therefore, bariatric surgery may be a reasonable therapeutic option for select patients.

In addition to its strengths, the study has limitations, the most important of which is a lack of comparator or control groups, making it challenging to ascertain the relative kidneyrelated benefits and risks of bariatric surgery. An additional limitation was the modest proportion of patients with existing CKD, which explained why so few participants progressed to ESRD. These limitations can be overcome by designing weight loss trials that include controls and enroll patients with established CKD. Missing data were another limitation, and they were related to patients either missing follow-up appointments or refusing blood draws. However, many of the missing visits in year 7 occurred as a result of the study being closed before the 7year anniversary date; therefore, these can be regarded as missing completely at random and less likely to introduce bias into our analyses. Other missing visits were treated as missing at

random and were appropriately accounted for by including predictors of missing data into the regression models. In addition, the analysis that used year 1 as baseline further reduced concerns about missing data, because there was 1 less year of potential missing data. Another limitation involved an abrupt and modest drift in cystatin C measurements early on in the study that was related to the manufacturer's calibrator method. However, the robustness of our results is shown by the fact that eGFR_{cr} (using rigorous calibration) did not qualitatively change our findings. Finally, although it is reasonable to assume that the predominant causes of CKD were diabetic kidney disease (because about one third of the population had diabetes) or obesityrelated glomerulopathy, the lack of kidney biopsy data prevents confirmation.

In conclusion, treatment of severe obesity with bariatric surgery was associated with improvements in CKD risk categories for up to 7 years in a large proportion of patients, especially those with moderate to high baseline risk. These findings support consideration of CKD risk in evaluation for bariatric surgery and further study of bariatric surgery as a treatment for high-risk obese patients with CKD.

CONCISE METHODS

Study Population

The study was made up of participants in the Longitudinal Assessment of Bariatric Surgery-2 (LABS-2) Study, an National Institutes of Health–sponsored, prospective, longitudinal, multicenter trial designed to study the long-term safety and efficacy of bariatric surgery and its effects on a variety of health-related parameters (trial registration: NCT00465829).²⁹ The LABS-2 Study participants were at least 18 years old and underwent firsttime bariatric procedures between March of 2006 and April of 2009.

Measures

Kidney disease was a prespecified end point of the LABS-2 trial, so serum creatinine, serum cystatin C, and the urine ACR were routinely measured. The study assessed CKD risk using the nomenclature system developed by the KDIGO consortium, which is the most recent internationally accepted criteria established for CKD.²⁶ The classification offers four CKD risk categories on the basis of a combination of eGFR_{cr-cys} and ACR: (1) low risk: an eGFR_{cr-cys} \geq 60 ml/min per 1.73 m² and an ACR<30 mg/g (this category includes patients without CKD);

(2) moderate risk: an eGFR_{cr-cys} between 45 and 59 ml/min per 1.73 m² and an ACR<30 mg/g or an eGFR_{cr-cvs} \geq 60 ml/min per 1.73 m² and an ACR between 30 and 300 mg/g; (3) high risk: an eGFR_{cr-cvs} between 30 and 44 ml/min per 1.73 m² and an ACR<30 mg/g, an eGFR_{cr-cys} between 45 and 59 ml/min per 1.73 m² and an ACR between 30 and 300 mg/g, or an eGFR \geq 60 ml/min per 1.73 m² and an ACR>300 mg/g; and (4) very high risk: eGFR_{cr-cys}<30 ml/min per 1.73 m² and an ACR<30 mg/g, an eGFR_{cr-cvs}<45 ml/min per 1.73 m² and an ACR between 30 and 300 mg/g, or an eGFR_{cr-cys}<60 ml/min per 1.73 m² and an ACR>300 mg/g. These categories predict the likelihood of having progressive CKD, AKI, or kidney failure and also closely correlate with cardiovascular and all-cause mortality risk.²⁶ The classification used the serum creatinine-based Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI) equation¹⁹ to estimate GFR and describes the kidney abnormalities as lasting at least 3 months. Our study used an updated version of the CKD-EPI equation²⁰ and did not verify chronicity.

Details on data collection, measurements, and definitions of diseases in the LABS-2 Study have previously been published (Supplemental Material).^{29–31}

Statistical Analyses

Baseline characteristics were summarized by CKD risk categories (low, moderate, high, and very high) among patients with complete information for baseline CKD risk (n=2144). Categorical data were summarized with frequencies and percentages, and continuous data were summarized with medians and interquartile ranges. Changes of CKD risk from baseline and year 1 were categorized as improvement (moving to a lower CKD risk category), no change (staying in the same risk category), or worsening (moving to a higher CKD risk category), and they were summarized with frequencies and percentages.

Ordinal logistic regression model was fitted to assess the association of baseline covariates with baseline CKD risk categories. Because there were relatively few participants at very high prognostic risk, for this analysis we combined high- and very highrisk groups to increase statistical power. We first tested if the proportional odds assumption that the odds of having high risk compared with those of having moderate or low CKD risk were the same as those of having moderate or higher risk compared with low risk was met. If so, a single odds ratio and 95% CI were reported. Otherwise, a generalized logistic regression model was fitted with lowest risk category as the reference, and odds ratios comparing higher-risk groups with the lowest-risk group were reported. Independent covariates with <0.2 significance level in the univariable analysis (*i.e.*, age, sex, BMI, HbA1c, serum creatinine, insulin resistance by homeostatic model assessment, presence of diabetes, presence of hypertension, and use of ACE inhibitors or ARBs) were included in the multivariable model, wherein a stepwise regression with a *P* value cutoff of 0.10 for both entry and removal procedures was used for covariate selection. Regardless of *P* value, covariates that, if removed, would significantly (>20%) change the coefficients of other covariates remained in the model.

Generalized linear mixed models were used for subsequent analyses to account for repeated measures and within-subject correlations. All models were adjusted for race, site, and baseline smoking status, because these variables were associated with missing follow-up data and needed to be in the model to produce accurate estimates of the odds ratios under missing at random assumption (Supplemental Table 9). The missing at random assumption was deemed reasonable by analyzing whether there were any systematic patterns that gave rise to incomplete data in CKD risk categories during follow-up. Among the participants missing CKD risk categories during follow-up versus not missing these data, the changing pattern of predicted probability of being in moderate or higher CKD risk group was not distributed differently over other time points after controlling for missing related baseline variables (i.e., race, site, and baseline smoking status) (Supplemental Table 10). This indicates that missing follow-up data were not related to outcome and should, therefore, not bias the results.

To investigate the trends in CKD risk categories over time, generalized ordinal logistic regression models stratified by baseline and year 1 risk category were used to estimate the proportions of changing status in the KDIGO risk level.

To evaluate the roles of baseline predictors for CKD risk after bariatric surgery, longitudinal analyses were performed using random intercept logistic regression models with risk categories as outcome (moderate or higher risk versus low risk) and visit (treated as categories), baseline covariates, follow-up weight loss, and their interaction with visit as covariates. Potential covariates and their interaction with visit were tested individually. Independent covariates with a P value <0.40 in univariable analyses (*i.e.*, age, sex, BMI, HbA1c, systolic BP, use of ACE inhibitors or ARBs, no private medical insurance, education, annual household income, surgery type, and percentage weight loss from baseline) were included in the adjusted model as well as corresponding interaction with follow-up that had *P* value <0.05. Partial *F* values with *P* value <0.20 from type 3 sum of squares were used to determine whether to keep covariates in the final model. RR ratios and 95% CIs are reported in the tables.

All reported *P* values are two sided, and *P* values < 0.05 were considered to be statistically significant.

ACKNOWLEDGMENTS

This clinical study was a cooperative agreement funded by the National Institute of Diabetes and Digestive and Kidney Diseases (NIDDK): grants U01 DK066557 (to Data Coordinating Center), U01-DK66667 (to Columbia in collaboration with Cornell University Medical Center Clinical Translational Research Center [CTRC] grant UL1-RR024996), U01-DK66568 (to the University of

Washington in collaboration with CTRC grant M01RR-00037), U01-DK66471 (to the Neuropsychiatric Research Institute), U01-DK66526 (to East Carolina University), U01-DK66585 (to the University of Pittsburgh Medical Center in collaboration with CTRC grant UL1-RR024153), and U01-DK66555 (to Oregon Health Sciences University).

The NIDDK scientists contributed to the design and conduct of the study, which included collection and management of data. The project scientist from the NIDDK served as a member of the steering committee along with the principal investigator from each clinical site and the data coordinating center. The data coordinating center housed all data during the study and performed data analyses according to a prespecified plan developed by the data coordinating center biostatistician and approved by the steering committee and the independent data and safety monitoring board. The decision to publish was made by the LABS-2 Study steering committee, with no restrictions imposed by the sponsor. As a coauthor (P.L.K.), an NIDDK scientist contributed to the interpretation of the data and preparation, review, or approval of the manuscript.

DISCLOSURES

None.

REFERENCES

- Courcoulas AP, Yanovski SZ, Bonds D, Eggerman TL, Horlick M, Staten MA, Arterburn DE: Long-term outcomes of bariatric surgery: A National Institutes of Health symposium. JAMA Surg 149: 1323–1329, 2014
- Sjöström L, Lindroos AK, Peltonen M, Torgerson J, Bouchard C, Carlsson B, Dahlgren S, Larsson B, Narbro K, Sjöström CD, Sullivan M, Wedel H; Swedish Obese Subjects Study Scientific Group: Lifestyle, diabetes, and cardiovascular risk factors 10 years after bariatric surgery. N Engl J Med 351: 2683–2693, 2004
- Schauer PR, Bhatt DL, Kirwan JP, Wolski K, Brethauer SA, Navaneethan SD, Aminian A, Pothier CE, Kim ES, Nissen SE, Kashyap SR; STAMPEDE Investigators: Bariatric surgery versus intensive medical therapy for diabetes–3-year outcomes. N Engl J Med 370: 2002–2013, 2014
- Courcoulas AP, Goodpaster BH, Eagleton JK, Belle SH, Kalarchian MA, Lang W, Toledo FG, Jakicic JM: Surgical vs medical treatments for type 2 diabetes mellitus: A randomized clinical trial. JAMA Surg 149: 707–715, 2014
- Courcoulas AP, Christian NJ, Belle SH, Berk PD, Flum DR, Garcia L, Horlick M, Kalarchian MA, King WC, Mitchell JE, Patterson EJ, Pender JR, Pomp A, Pories WJ, Thirlby RC, Yanovski SZ, Wolfe BM; Longitudinal Assessment of Bariatric Surgery (LABS) Consortium: Weight change and health outcomes at 3 years after bariatric surgery among individuals with severe obesity. JAMA 310: 2416–2425, 2013
- Friedman AN, Wolfe B: Is bariatric surgery an effective treatment for type II diabetic kidney disease? Clin J Am Soc Nephrol 11: 528–535, 2016
- D'Agati VD, Chagnac A, de Vries AP, Levi M, Porrini E, Herman-Edelstein M, Praga M: Obesity-related glomerulopathy: Clinical and pathologic characteristics and pathogenesis. *Nat Rev Nephrol* 12: 453–471, 2016
- Vivante A, Golan E, Tzur D, Leiba A, Tirosh A, Skorecki K, Calderon-Margalit R: Body mass index in 1.2 million adolescents and risk for endstage renal disease. Arch Intern Med 172: 1644–1650, 2012

- Wang Y, Chen X, Song Y, Caballero B, Cheskin LJ: Association between obesity and kidney disease: A systematic review and meta-analysis. *Kidney Int* 73: 19–33, 2008
- Friedman A: Implications and management of obesity in kidney disease. In: Nutrition in Kidney Disesae, edited by Byham-Gray L, Burrows J, Chertow G, New York, Springer, 2014, pp 147–169
- Chang AR, Grams ME, Navaneethan SD: Bariatric surgery and kidneyrelated outcomes. Kidney Int Rep 2: 261–270, 2017
- Friedman AN, Moe S, Fadel WF, Inman M, Mattar SG, Shihabi Z, Quinney SK: Predicting the glomerular filtration rate in bariatric surgery patients. Am J Nephrol 39: 8–15, 2014
- Chagnac A, Weinstein T, Herman M, Hirsh J, Gafter U, Ori Y: The effects of weight loss on renal function in patients with severe obesity. J Am Soc Nephrol 14: 1480–1486, 2003
- Li K, Zou J, Ye Z, Di J, Han X, Zhang H, Liu W, Ren Q, Zhang P: Effects of bariatric surgery on renal function in obese patients: A systematic review and meta analysis. *PLoS One* 11: e0163907, 2016
- 15. Navaneethan SD, Yehnert H: Bariatric surgery and progression of chronic kidney disease. *Surg Obes Relat Dis* 5: 662–665, 2009
- Imam TH, Fischer H, Jing B, Burchette R, Henry S, DeRose SF, Coleman KJ: Estimated GFR before and after bariatric surgery in CKD. Am J Kidney Dis 69: 380–388, 2017
- Chang AR, Chen Y, Still C, Wood GC, Kirchner HL, Lewis M, Kramer H, Hartle JE, Carey D, Appel LJ, Grams ME: Bariatric surgery is associated with improvement in kidney outcomes. *Kidney Int* 90: 164–171, 2016
- Nehus EJ, Khoury JC, Inge TH, Xiao N, Jenkins TM, Moxey-Mims MM, Mitsnefes MM: Kidney outcomes three years after bariatric surgery in severely obese adolescents. *Kidney Int* 91: 451–458, 2017
- Levey AS, Stevens LA, Schmid CH, Zhang YL, Castro AF 3rd, Feldman HI, Kusek JW, Eggers P, Van Lente F, Greene T, Coresh J; CKD-EPI (Chronic Kidney Disease Epidemiology Collaboration): A new equation to estimate glomerular filtration rate. *Ann Intern Med* 150: 604–612, 2009
- Inker LA, Schmid CH, Tighiouart H, Eckfeldt JH, Feldman HI, Greene T, Kusek JW, Manzi J, Van Lente F, Zhang YL, Coresh J, Levey AS; CKD-EPI Investigators: Estimating glomerular filtration rate from serum creatinine and cystatin C. N Engl J Med 367: 20–29, 2012
- Carey DG, Pliego GJ, Raymond RL: Body composition and metabolic changes following bariatric surgery: Effects on fat mass, lean mass and basal metabolic rate: Six months to one-year follow-up. Obes Surg 16: 1602–1608, 2006
- 22. Levey AS, Inker LA, Coresh J: GFR estimation: From physiology to public health. *Am J Kidney Dis* 63: 820–834, 2014
- Mallamaci F, Ruggenenti P, Perna A, Leonardis D, Tripepi R, Tripepi G, Remuzzi G, Zoccali C; REIN Study Group: ACE inhibition is renoprotective among obese patients with proteinuria. J Am Soc Nephrol 22: 1122–1128, 2011
- Neff KJ, Elliott JA, Corteville C, Abegg K, Boza C, Lutz TA, Docherty NG, le Roux CW: Effect of Roux-en-Y gastric bypass and diet-induced weight loss on diabetic kidney disease in the Zucker diabetic fatty rat. Surg Obes Relat Dis 13: 21–27, 2016
- MacLaughlin HL, Hall WL, Patel AG, Blacklock RM, Swift PA, Phanish MK, Dew T, Chowdhury P, Sanders TAB, Macdougall IC: Weight loss, adipokines, and quality of life after sleeve gastrectomy in obese patients with stages 3-4 CKD: A randomized controlled pilot study. Am J Kidney Dis 64: 660–663, 2014
- Kidney Disease: Improving Global Outcomes (KDIGO) CKD Work Group: KDIGO 2012 clinical practice guideline for the evaluation and management of chronic kidney disease. *Kidney Int Suppl* 3: S1–S150, 2013
- Turgeon NA, Perez S, Mondestin M, Davis SS, Lin E, Tata S, Kirk AD, Larsen CP, Pearson TC, Sweeney JF: The impact of renal function on outcomes of bariatric surgery. J Am Soc Nephrol 23: 885–894, 2012

CLINICAL EPIDEMIOLOGY www.jasn.org

- US Renal Data System: USRDS 2016 Annual Data Report: Atlas of End-Stage Renal Disease in the United States, Bethesda, MD, National Institutes of Health, National Institute of Diabetes and Digestive and Kidney Diseases, 2016
- Belle SH, Berk PD, Chapman WH, Christian NJ, Courcoulas AP, Dakin GF, Flum DR, Horlick M, King WC, McCloskey CA, Mitchell JE, Patterson EJ, Pender JR, Steffen KJ, Thirlby RC, Wolfe BM, Yanovski SZ; LABS Consortium: Baseline characteristics of participants in the Longitudinal Assessment of Bariatric Surgery-2 (LABS-2) study. *Surg Obes Relat Dis* 9: 926–935, 2013
- Belle SH, Berk PD, Courcoulas AP, Flum DR, Miles CW, Mitchell JE, Pories WJ, Wolfe BM, Yanovski SZ; Longitudinal Assessment of Bariatric Surgery Consortium Writing Group: Safety and efficacy of bariatric

surgery: Longitudinal assessment of bariatric surgery. *Surg Obes Relat Dis* 3: 116–126, 2007

 Buse JB, Caprio S, Cefalu WT, Ceriello A, Del Prato S, Inzucchi SE, McLaughlin S, Phillips GL 2nd, Robertson RP, Rubino F, Kahn R, Kirkman MS: How do we define cure of diabetes? *Diabetes Care* 32: 2133–2135, 2009

This article contains supplemental material online at http://jasn.asnjournals. org/lookup/suppl/doi:10.1681/ASN.2017060707/-/DCSupplemental.

See related perspective, "Bariatric Surgery and Kidney Health," on pages 1085–1086.